Author's response to reviews

Title: Sagittal Jaw Position in relation to Body Posture in Adult Humans - a Rasterstereographic Study.

Authors:

Carsten Lippold (lippold@uni-muenster.de)
Gholamreza Danesh (siamak@uni-muenster.de)
Markus Schilgen (markus.schilgen@manuellemedizin.de)
Burkhard Drerup (drerup@uni-muenster.de)
Lars Hackenberg (hackenberg@uni-muenster.de)

Version: 4 Date: 20 December 2005

Author's response to reviews: see over
Dear Editorial Team,

we want to thank you for completing the peer review process of our manuscript “Sagittal Jaw Position in relation to Body Posture in Adult Humans - a Rasterstereographic Study”. According to Professor Ferrario we changed the captions to table #1 into “class I, II and III”.

It was the first time for us publishing with BMC - and we were very content with the reviewing process. So again thank you much!

Sincerely

Carsten Lippold, Dr. med. dent.

Department of Orthodontics,
Center of Dental, Oral and Maxillofacial Diseases
University of Münster,
Waldeyerstraße 30
D-48149 Münster
Germany

8196173427847634